Author's response to reviews

Title: Priority setting and health policy and systems research

Authors:

Michael K Ranson (ransonm@who.int)
Sara C Bennett (saracbennett@gmail.com)

Version: 4 Date: 30 November 2009

Author's response to reviews:

I have responded to all of the changes requested below. Please do note, References 19 and 20 remain "in Press". Please do not hesitate to contact me, should you require any additional information.

Formatting changes to be made:

---------------------------------------------
Major revisions (we require the authors to make these changes)

Competing interests - Please include a `Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write `The authors declare that they have no competing interests'.

Authors' contributions - Please place the Authors' contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

Please use the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Acknowledgements - If you have any acknowledgments, please place an Acknowledgements section at the end of the manuscript text, directly preceding the Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship.

References 19 and 20 in Press; have these references been published? If so, please include full citation details.

Minor revisions (we can make these changes for the authors, however it will speed publication of your article if you make these revisions along with the major
revisions above)

Title page: please remove the words 'author list' and 'affiliations'.

Manuscript format: please remove all section numbering.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Thank you